

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL106585

Key Points:

- Observed tropical Atlantic forced pacemaker experiments produce La Niña events, but not El Niño events, in unison with observations
- A warm Atlantic had a greater chance of leading to a La Niña event if the tropical Pacific was initially in neutral conditions
- There was a tropical-wide response to a warm Atlantic in the model, affecting western Pacific wind. This was not seen for a cold Atlantic

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

P. van Rensch, peter.vanrensch@monash.edu

Citation

van Rensch, P., McGregor, S., Dommenget, D., Bi, D., & Liguori, G. (2024). The tropical Atlantic's asymmetric impact on the El Niño-Southern Oscillation. *Geophysical Research Letters*, *51*, e2023GL106585. https://doi.org/10. 1029/2023GL106585

Received 27 SEP 2023 Accepted 14 FEB 2024

© 2024. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

The Tropical Atlantic's Asymmetric Impact on the El Niño-Southern Oscillation

Peter van Rensch^{1,2} ^(D), Shayne McGregor^{1,2} ^(D), Dietmar Dommenget^{1,2} ^(D), Daohua Bi³ ^(D), and Giovanni Liguori^{1,4} ^(D)

¹School of Earth, Atmosphere and Environment, Monash University, Clayton, VIC, Australia, ²Centre of Excellence for Climate Extremes, School of Earth Atmosphere and Environment, Monash University, Clayton, VIC, Australia, ³CSIRO Environment, Aspendale, VIC, Australia, ⁴Department of Biological, Geological, and Environmental Sciences (BIGEA), University of Bologna, Bologna, Italy

Abstract Using observations and Atlantic forced coupled model simulations, we show evidence for an asymmetry in the link between beginning of year tropical Atlantic sea surface temperature anomalies (SSTAs) and end of the year El Niño-Southern Oscillation events. We find a greater tendency for warm Atlantic SSTAs to lead to a La Niña than for cold anomalies to lead to El Niño. The model experiments showed that the Atlantic had a greater chance to force the tropical Pacific if the Pacific was initially in a neutral state. In the model, a warm Atlantic from March–May was able to produce an atmospheric response leading to easterly wind anomalies in the western Pacific. This in turn induces a subsurface oceanic response, leading to La Niña at the end of the year. The atmospheric response does not occur for a cold Atlantic, leading to no impacts in the Pacific.

Plain Language Summary It has been found that the tropical Atlantic can have an impact on an El Niño or La Niña events that form in the tropical Pacific at the end of the year. We show using observations and targeted model experiments that a warmer than normal Atlantic from March-May can lead to a La Niña, but a colder than normal Atlantic does not lead to an El Niño. The model shows a strong atmospheric response to warm Atlantic surface temperature, leading to wind changes in the Pacific that aid La Niña formation. This strong atmospheric response does not occur for cold Atlantic temperatures leading to no changes in the Pacific. The model also shows that the Atlantic has a greater chance of impacting the Pacific at the end of the year if the Pacific is already in near average conditions at the beginning of the year. These results may improve forecasts of La Niña events and may help in understanding past changes in the Atlantic-Pacific relationship.

1. Introduction

The El Niño-Southern Oscillation (ENSO) has a large influence on global weather, leading to significant human impacts, ranging from economic (Callahan & Mankin, 2023) and agricultural (Iizumi et al., 2014) to emergency management (Guimarães Nobre et al., 2019). Accurate forecasting of ENSO events provides a vital warning for these sectors, meaning any improvements to the forecast could have an appreciable human benefit. One avenue to improve forecasts is through understanding the influence of other tropical ocean basins on ENSO development in the Pacific (Cai et al., 2019). Studies have shown that the tropical Atlantic can have an impact on ENSO events, ranging from causing ENSO events (Ham et al., 2013; Polo et al., 2015) to modulating ENSO events (Chikamoto et al., 2020; Ding et al., 2012; Richter et al., 2021; Rodríguez-Fonseca et al., 2009).

The current leading theory for the Atlantic to impact the Pacific is associated with a Gill-type atmosphere response (Gill, 1980) originating in the Atlantic (Ding et al., 2012; Ham et al., 2013; Jiang & Li, 2021; Polo et al., 2015; Rodríguez-Fonseca et al., 2009). Sea surface temperature anomalies lead to a change in convection and upper-level heating, which is dissipated through the tropics through fast-moving eastward propagating Kelvin waves and slow, westward moving Rossby waves. The two waves converge over the Pacific leading to a changed circulation and surface wind anomalies. The Pacific wind anomalies influences the ocean subsurface through the Bjerknes feedback, potentially growing to an ENSO event.

It is also possible that the Atlantic to Pacific connection is the result of a spurious relationship between ENSO and the Atlantic (Zhang et al., 2021). The influence of the Atlantic on the Pacific could be a statistical artifact resulting from the combination of autocorrelation within the Pacific due to internal recharge oscillator dynamics, and ENSO variability also influencing Atlantic SSTAs. This suggests it will be challenging to investigate Atlantic—

Pacific interactions using observations alone. However, studies using coupled models forced by Atlantic SSTs report changes in the modeled ENSO (Bi et al., 2022; Chikamoto et al., 2020; Ding et al., 2012; Polo et al., 2015), suggesting there is a pathway for the Atlantic to affect the Pacific in the model space.

It has been shown that cold SSTAs produce a weaker magnitude global atmospheric response compared to their opposite warm SSTAs (Williams et al., 2023). Whether this asymmetry in response can occur in the Atlantic has begun to be explored. A modeling study found that a cold Atlantic exerts a slightly stronger magnitude wind response over the Pacific than for a warm Atlantic (Richter et al., 2021). This raises the question whether there is an asymmetry in the Pacific response to cold and warm Atlantic SSTs and if it can be detected in observations and in other general circulation models.

This paper addresses this issue by first detailing the methods and data sets used (Section 2), then explore the asymmetry in observations (Section 3), and Atlantic pacemaker experiments (Section 4). We then explore the dependence of the Pacific initial state to the ability for the Atlantic to have an impact (Section 5), and propose a mechanism for the asymmetry (Section 6).

2. Methods

2.1. Observed Analysis

For SST observations we used HadISST (Rayner, 2003) and ERSSTv3b (Smith et al., 2008) depending on the model used for comparison (see Section 2.4). We used the time period from December 1978 to February 2014, and we linearly detrend the data before performing regressions.

2.2. Definition of Events

We used the mean (0 anomaly) value to split between warm or cold SSTA groups. For composites, La Niña and El Niño events were defined as DJF Niño3.4 less than -1 and greater than 1 standard deviation, respectively. We used all members of the model experiment to calculate the model standard deviation.

2.3. Models Utilized

Two versions of the Australian Community Climate and Earth System Simulator (ACCESS) were used. ACCESS CM2 (Bi et al., 2020), which is the latest version submitted to CMIP6 (Coupled Model Intercomparison Project Phase 6), and ACCESS 1.0 (Bi et al., 2013), which is an older version submitted to CMIP5 (see Text S1 in Supporting Information S1 for details).

The models share a similar base, but the fully coupled control simulations have different dynamics in their representation of ENSO and Atlantic—Pacific relationship. The correlation between MAM Atlantic basin wide mean $(10^{\circ}N-10^{\circ}S, 60^{\circ}W-15^{\circ}E)$ and the following DJF Pacific basin wide mean $(10^{\circ}N-10^{\circ}S, 130^{\circ}E-80^{\circ}W)$ in an ACCESS 1.0 control run is -0.14, weaker than for all five ACCESS CM2 control runs (ranging from -0.15 to -0.48). Both models are weaker than that observed in HadISST (-0.54) and ERSST (-0.53) over the same period from 1979 to 2014. The fully coupled ACCESS CM2 model had the strongest North Tropical Atlantic—ENSO relationship of all CMIP6 models analyzed in Zhang et al. (2021), making it the closest to the relationship observed from 1990 to 2019.

There are numerous differences in the modeled representation of ENSO statistics and dynamics between AC-CESS 1.0 and ACCESS CM2 (Planton et al., 2021). For example, whereas ACCESS CM2 produces a better ENSO pattern close to that observed, ACCESS 1.0 produces a more accurate ENSO lifecycle than ACCESS CM2 (Planton et al., 2021).

2.4. Pacemaker Experiment Details

We analyzed results from pacemaker experiments where the model Atlantic SSTs were prescribed to observed SSTs, while the atmosphere and remaining oceans were free to evolve. The two ACCESS model versions used two different Atlantic pacemaker experiment configurations, as they were originally developed independently. The three key differences are the Atlantic meridional extent used as a pacemaker (ACCESS 1.0: 30°S–30°N; ACCESS CM2: 20°S–20°N), the observational data set used to restore the observed SSTs (ACCESS 1.0: ERSSTv3b; ACCESS CM2: HadISST), and the method used to restore the SSTs (ACCESS 1.0: SSTs nudged to

observations using a 3-day restoring timescale—for further details, see Purich et al. (2021); ACCESS CM2: SSTs are bypassed and completely replaced with observations—see Bi et al. (2022)). The integration time for ACCESS 1.0 and ACCESS CM2 was from 1961–2014 and 1970–2014, respectively, but we only used data from 1979 to 2014. Five runs were available for ACCESS CM2 and 10 runs for ACCESS 1.0.

2.5. Statistical Significance

Statistical significance for the regression analysis was calculated using the Student's *t*-test with a confidence interval of 0.05 p-value or less. For the statistical significance in the Venn diagrams, we used Theiler surrogates to produce random timeseries of the observed DJF Niño3.4 with similar spectral properties (Prichard & Theiler, 1994; Theiler et al., 1992), using the phaseran tool (Khider et al., 2022). We then perform the same analysis 1000 times to calculate the 5%–95% statistical significance bounds.

3. Observations Show a Stronger Relationship Between MAM Atlantic and La Niña Than El Niño

First, using observations, we detect if regions of the Atlantic SST in MAM are coherent with Niño3.4 in the following DJF, showing many regions of significant correlation (Figure 1a). We focus on the western tropical Atlantic (WTA), due to its strong coherence and tropical location. A scatter plot shows an asymmetry in this relationship, where warm (positive) anomalies in this region favors La Niña conditions, but cold (negative) anomalies do not favor El Niño conditions (Figure 1b).

We can perform the same analysis as Figure 1b for each grid point of the SST data set, whereby for each grid point we select the years of negative anomalies of MAM SSTs and correlate with the corresponding year of DJF Niño3.4 (Figure 1c), likewise for positive anomalies (Figure 1d). This reveals the regions of anomalously cold or warm MAM SSTs that may lead to a change in end of year DJF Niño3.4.

Cold tropical Pacific DJF SSTA (La Niña) were affected by earlier MAM SSTAs in more regions of the globe than warm tropical Pacific DJF SSTA (El Niño) (compare blue regions to red regions in Figures 1c and 1d). Cold MAM SSTAs in the central tropical Pacific and southwest tropical Indian Ocean were related to La Niña conditions later in DJF (Figure 1c). The central tropical Pacific signal is consistent with the tendency for La Niña to extend over multiple years (Okumura & Deser, 2010).

Positive MAM SSTAs show many well-defined regions that were related to end of year La Niña conditions (Figure 1d). The western tropical Atlantic showed a high coherence, along with the eastern off-equatorial Indian Ocean and western north Pacific. A patchy region in the eastern tropical Pacific could be remnants from a preceding El Niño (Figure 1d), consistent with the tendency for El Niño to be followed by La Niña (Okumura & Deser, 2010). These findings can be reproduced using another observational data set (Figure S1 in Supporting Information S1), however, the possibility that these features are an artifact similar in nature to that described by Zhang et al. (2021) (i.e., a confounding of ENSO autocorrelation and ENSO teleconnections) cannot be ruled out.

4. La Niña in Atlantic Pacemakers Are More Likely to Occur in Unison With Observations Than El Niño

To overcome the effects of autocorrelation and Atlantic teleconnection within the observed Pacific variability, we examined the results of pacemaker experiments. In these experiments the Atlantic SSTs were prescribed to observed SSTs, while the atmosphere and remaining oceans were free to evolve. This means that an El Niño or La Niña was not able to change the Atlantic SSTs, breaking the Pacific-Atlantic teleconnection route described by Zhang et al. (2021).

The model ensemble mean Niño3.4 region SSTA possessed a synchronisation with the observed Niño3.4 region SSTA. In both climate models, this synchronisation was stronger for cold Pacific SSTA only, indicating that the modeled and observed La Niña events tend to coincide (Figure 2a). This suggests that the Atlantic had a more robust impact on the formation of La Niña events than El Niño events in these models.

Focusing on the spread between the individual ensemble members in any given year can also give an indication of the statistical significance of this interbasin connectivity and any other factors affecting this relationship. For instance, there were six La Niña events in the observations during 1979–2014–which equates to 30 possible years





Figure 1. (a) Correlation of DJF Niño3.4 with preceding MAM SSTAs. Black contours show 95% significant values using Student's *t*-test. Gray dotted boxes show the Niño3.4 ($5^{\circ}N-5^{\circ}S$, $170^{\circ}W-120^{\circ}W$) and WTA ($10^{\circ}N-10^{\circ}S$, $60^{\circ}W-15^{\circ}W$) regions. (b) Observed scatterplot of DJF Niño 3.4 versus the preceding MAM WTA SSTA with linear regressions using all, only cold and only warm Atlantic SSTs. (c) Correlations with DJF Niño3.4 using only negative MAM SSTAs at each grid point. (d) As (c) but using only positive MAM SSTAs. We multiply (c) by -1 so the blue shades show regions related to end-of-year warm Niño3.4. This figure only used data from HadISST.

in the five-member ensemble of ACCESS CM2 that could coincide with the observations (six observed La Niña times five realizations).

Of the 31 La Niña events that occurred in the ACCESS CM2 ensemble during 1979–2014, there were 13 instances in which there was both a La Niña in the model and a La Niña in the observations (Figure 2b). This is outside the range expected by random chance defined by the 95th confidence level (see Section 2.4). Further to this, when the ACCESS CM2 pacemaker ensemble exhibited a La Niña, there was a 42% chance that it occurred in the same year as observed. Whereas the number of El Niño events in the ACCESS CM2 pacemaker that cooccurred with the observations could be explained by random chance (Figure 2b, bottom row).

Carrying out the same analysis for ACCESS 1.0 pacemaker, we find that the co-occurrence of La Niña events only occurred 15% of the time—which is within the possibility of random chance (Figure 2c). This weak result could be expected given the ACCESS 1.0 control exhibited a limited interbasin connection than that seen in both observations and ACCESS CM2 (see Section 2.3), with SSTAs leading up to the formation of ENSO events largely limited to the Pacific Ocean (Figure S1 in Supporting Information S1). ENSO in ACCESS 1.0 is less biennial compared to ACCESS CM2, which may also be a factor in the Atlantic-Pacific interaction (e.g., Zhang et al., 2021).





Figure 2. Synchronisation of Atlantic pacemaker to observations in the Pacific. (a) Multi-run mean of Atlantic pacemaker DJF Niño3.4 versus the observed DJF Niño3.4 from 1979 to 2014. ACCESS CM2 and HadISST are blue, ACCESS 1.0 and ERSST are orange. Regressions for both positive and negative observed Niño3.4 are shown in solid and dashed lines, respectively (see legend). All timeseries were linearly detrended. (b) Venn diagram showing the total number of ACCESS CM2 pacemaker years in all runs that correspond with a La Niña in the model and/or HadISST (events with Niño3.4 standard deviation of -1 or less). The size of the circles are proportional to the sample size. (c) As for (b) but using ACCESS 1.0 pacemaker years and ERSSTv3b for observations. The bottom row shows the same but for El Niño events. The range shown in the Successful group is the 5th–95th confidence interval (see Section 2.5), meaning counts outside this range are statistically significant.

5. Pacific Initial State May Determine if a Warm Atlantic Can Lead to La Niña

The ACCESS CM2 ensemble showed that while a warm Atlantic on average forces a La Niña-like Pacific response, it does not force a La Niña on all occasions. Here we used composite analysis to explore possible reasons why modeled La Niña events successfully occurred at the same time as observations, why they occurred independently of the observations, and why they failed to occur at the same time as observations, as characterized in Figure 2b (Figure 3).



Geophysical Research Letters



Figure 3. Composite mean of SSTAs for the (top row) successful, (middle row) independent, (bottom row) failed La Niña for ACCESS CM2, as defined in Figure 2b. Each column from left to right shows the time progression leading to the event: DJF and MAM before the event with the event occurring in DJF.

The composite analysis reveals that successful La Niña events show a widespread warm SSTA in the Atlantic in the preceding DJF that largely persists into MAM (Figures 3a and 3b), indicating the Atlantic was a strong driver for the La Niña synchronisation with observations. Further to this, there is an indication that the equatorial Pacific was ENSO neutral at the beginning of the year (Figure 3a), with robust neutral conditions forming by April (Figure S2a in Supporting Information S1). This follows a weak equatorial Pacific subsurface temperature anomaly during this time (sea surface height anomaly, Figure S2a in Supporting Information S1), indicating the Pacific may have been more open to external influences during MAM.

Independent La Niña events did not tend to have an Atlantic SSTA during the lead up, indicating that Atlantic SSTAs had little systematic impact on these events (Figures 3d and 3e). Rather, independent La Niña events tended to have a strong El Niño anomaly in the preceding DJF (which discharged heat throughout the year leading to La Niña, e.g. Jin (1997), Figure S2b in Supporting Information S1) and a warm Indian Ocean in MAM, both of which were not widely seen in the successful and failed (see below) groups. While ACCESS CM2 has improved in other ENSO metrics, it displays a higher frequency ENSO cycle compared to observations and most other CMIP5 and CMIP6 models (Planton et al., 2021). Thus, the evolution of SSTA in the independent La Niña group appears consistent with the typical ENSO evolution of the fully coupled model, and in terms of the El Niño transition to La Niña, the observations (e.g., Okumura & Deser, 2010). Further to this, a warm Indian Ocean has also been highlighted as a precursor for La Niña events (e.g., Izumo et al., 2016), it is possible this may play a role in the development and/or amplification of the event in this model.

Like the successful La Niña events, failed La Niña events also had a persistent warm SSTA in the Atlantic in MAM, but these events also displayed a La Niña-like anomaly in the preceding DJF and MAM (Figures 3g and 3h) and a Pacific subsurface that was recharging heat since the previous year (Figure S2c in Supporting Information S1). The tendency in the Pacific subsurface was to warm, potentially to prime an El Niño event later in the event year following ENSO Recharge Oscillator dynamics (Jin, 1997), and also the biennial ENSO tendency of





Figure 4. (a) Regressions of (top) MAM 200 hPa temperature and winds, (center) MAM vertical temperature and winds averaged 10N–10S and (bottom) JJA sea surface height and MAM surface winds onto positive values of MAM western tropical Atlantic SSTA. Green contours indicate 0.05 p-values for the regression coefficient. Vertical velocity is multiplied by 1000. (b) As (a), but for negative values of MAM WTA SSTA; we multiply the negative WTA regressions by -1 for easier comparison to positive WTA regressions in (a).

this model (Bi et al., 2022). Potentially, the warm Atlantic SSTs were not enough to maintain a La Niña event, but may have prevented the transition from a La Niña to El Niño, which is typical in this model, however, there is a large range to these values (Figure S2c in Supporting Information S1). This could also be another interpretation of the results of Bi et al. (2022), where not only a more realistic Atlantic mean state improved ENSO periodicity, the suppression of these El Niño events also reduced the ENSO frequency.

The ACCESS 1.0 pacemaker showed similar progressions within the groups, despite not exhibiting a significant successful La Niña count (Figure 2c; Figure S3 in Supporting Information S1). The successful and failed groups showed a warm Atlantic in the build-up (Figures S3a, S3b, S3g, and S3h in Supporting Information S1). The failed group tended to show La Niña conditions in the year prior to the event (Figure S3g in Supporting Information S1). Whereas the independent La Niña show a weak indication of an El Niño occurring the year before (Figure S3d in Supporting Information S1). Furthermore, the El Niño case for ACCESS CM2 does not show a similar progression as the La Niña case (Figure S4 in Supporting Information S1), supporting the hypothesis that the Atlantic preferentially forces La Niña events than El Niño events.

6. Mechanism for an Asymmetric Response From Atlantic SSTAs

Our results show the Atlantic has the capacity to influence the formation of La Niña events in the Pacific Ocean, but less so for El Niño events. Here we explore a possible mechanism in the ACCESS CM2 pacemaker ensemble mean for a warm Atlantic to influence the Pacific, that was not seen for a cold Atlantic. We analyze atmospheric fields in MAM and sea surface height in JJA (as a metric for subsurface heat content) by regressing onto both positive and negative MAM WTA (Figure 4). The importance of Atlantic SSTA in MAM or JJA seasons to force the Pacific is still under debate (e.g., Jiang et al., 2022), we show MAM as we can begin to see anomalies that grow into JJA, but our pacemaker experiments were forced by the observed Atlantic SSTs in every month.

Warm MAM WTA SSTAs were consistent with a Gill-type response (Gill, 1980), characterized by a Rossby wave-like pattern with anticyclonic rotation at about 20° latitude in both hemispheres, and a Kelvin wave-like

19448007, 2024, 4, Downloaded

schu

loi/10.1029/2023GL106585 by Area Sistemi Dipart & Docum

Library on [10/04/2024]. See the Terms

pattern with warm upper-level temperature anomalies extending eastward along the equator (Figure 4a). The eastward extension appeared to converge over the western Pacific with a warm anomaly extending west from the Atlantic. Subsidence was enhanced where these signals converge leading to easterly wind anomalies at the surface in the western Pacific. These easterly wind anomalies induced a cool ocean subsurface temperature response that propagated to the east leading into JJA, activating the Bjerknes feedback and resulting in a La Niña signal at the end of the year (Figure 4a).

On the other hand, the cold MAM WTA SSTA had a limited tropospheric temperature response (Figure 4b), consistent with the results of Williams et al. (2023) which showed a weaker convection response for cold SST anomalies. The Rossby wave-like patterns straddling the equator were detected for the cold WTA, but the Kelvin wave-like eastward extension was not evident compared to the warm WTA case (cf. Figures 4a and 4b). The lack of the eastward extension appears responsible for the lack of response over the Pacific (Figure 4b), but why this did not occur is not known. One potential avenue to explore is the sensitivity of the height of the tropospheric temperature anomaly over the equatorial Atlantic for Kelvin wave-like responses (cf. Figures 4a and 4b).

7. Summary and Discussion

In this study we utilized observations and models to provide evidence that anomalously cold and warm Atlantic SSTAs exhibit an asymmetric relationship with tropical Pacific SSTAs. This suggests that there is potential for the Atlantic to be used to improve La Niña forecast, but not for El Niño forecast. The Atlantic had a greater chance of influencing La Niña events if the Pacific was in a neutral state in MAM, and the tropical Atlantic had a widespread warm pattern. The asymmetry from the Atlantic SSTAs was due to the lack of an upper-level atmospheric response for cool SSTAs, therefore leading to no wind anomalies at the surface in the western Pacific, whereas this was evident for warm Atlantic SSTAs.

Our observed and modeled results also showed an asymmetry in the influence of the Indian Ocean on the Pacific, where a warm Indian Ocean had a greater signal in forcing La Niña events than El Niño. The influence of the Indian Ocean on ENSO has been well documented in the literature (Cai et al., 2019; Izumo et al., 2016), but could be explored further using similar methodology to that showed here.

In the last two decades there has been a tendency for a warming Atlantic and a cooling Pacific (Cai et al., 2019; Li et al., 2016; McGregor et al., 2014; Naha et al., 2023; Ruprich-Robert et al., 2017, 2021). A warming Atlantic and an asymmetric relationship between the Atlantic and Pacific may lead to a cooling Pacific due to the occurrence of La Niña events, while still allowing for the Pacific to exhibit some autonomy to cause El Niño events, for example, the 2015–2016 event.

Subsequent studies could focus on asymmetries not only in the developing phase of ENSO, but also in the decaying phase (e.g., Richter et al., 2023). However, there remains some uncertainty in the precise mechanism for which the Atlantic can influence the Pacific in the developing phase alone. Model biases in the Atlantic mean state (McGregor et al., 2018) and the response to the Atlantic (Ruprich-Robert et al., 2021) likely hinders this pursuit. The effects of ENSO autocorrelation and teleconnection also cannot be ignored in the observations (e.g., Zhang et al., 2021), therefore a diverse set of analyses in numerous modeling configurations are likely required.

Data Availability Statement

HadISST version 1.1 (Rayner, 2003) and ERSST version 3b (Smith et al., 2008) was used for observed SST. ACCESS 1.0 (Bi et al., 2013) and ACCESS CM2 (Bi et al., 2020) were used to produce the model results. For analysis and plotting we used xarray version 2023.4.2 (Hoyer et al., 2023), matplotlib version 3.5.3 (Hunter, 2007), numpy version 1.23.5 (Harris et al., 2020), PyleoClim version 0.7.2 (Khider et al., 2022).

References

Bi, D., Dix, M., Marsland, S., O'Farrell, S., Rashid, H., Uotila, P., et al. (2013). The ACCESS coupled model: Description, control climate and evaluation [Software]. Australian Meteorological and Oceanographic Journal, 63, 41–64. https://doi.org/10.22499/2.6301.004

Bi, D., Dix, M., Marsland, S., O'Farrell, S., Sullivan, A., Bodman, R., et al. (2020). Configuration and spin-up of ACCESS-CM2, the new generation Australian Community Climate and Earth System Simulator Coupled Model [Software]. Journal of Southern Hemisphere Earth Systems Science, 70, 225–251. https://doi.org/10.1071/ES19040

Acknowledgments

The authors acknowledge the comments of two anonymous reviewers who greatly improved the quality of this paper. This work was supported by the Australian Research Council through Grants DP200102329 and CE170100023. This work was partly supported by the National Environmental Science Program (NESP) Climate Systems Hub project CS2.11. This research was undertaken using resources from the National Computational Infrastructure (NCI Australia). Open access publishing facilitated by Monash University, as part of the Wiley - Monash University agreement via the Council of Australian University Librarians.

on Wiley Online Librar

°A

are

applicable Creati

- Bi, D., Wang, G., Cai, W., Santoso, A., Sullivan, A., Ng, B., & Jia, F. (2022). Improved simulation of ENSO variability through feedback from the equatorial Atlantic in a pacemaker experiment. *Geophysical Research Letters*, 49(2), e2021GL096887. https://doi.org/10.1029/2021GL096887 Cai, W., Wu, L., Lengaigne, M., Li, T., McGregor, S., Kug, J.-S., et al. (2019). Pantropical climate interactions. *Science*, 363(6430), eaav4236. https://doi.org/10.1126/science.aav4236
- Callahan, C. W., & Mankin, J. S. (2023). Persistent effect of El Niño on global economic growth. *Science*, 380(6649), 1064–1069. https://doi.org/ 10.1126/science.adf2983
- Chikamoto, Y., Johnson, Z. F., Wang, S.-Y. S., McPhaden, M. J., & Mochizuki, T. (2020). El Niño–Southern Oscillation evolution modulated by Atlantic forcing. *Journal of Geophysical Research: Oceans*, 125(8), e2020JC016318. https://doi.org/10.1029/2020JC016318
- Ding, H., Keenlyside, N. S., & Latif, M. (2012). Impact of the equatorial Atlantic on the El Niño Southern Oscillation. *Climate Dynamics*, 38(9–10), 1965–1972. https://doi.org/10.1007/s00382-011-1097-y
- Gill, A. E. (1980). Some simple solutions for heat-induced tropical circulation. Quarterly Journal of the Royal Meteorological Society, 106(449), 447–462. https://doi.org/10.1002/qj.49710644905
- Guimarães Nobre, G., Muis, S., Veldkamp, T. I. E., & Ward, P. J. (2019). Achieving the reduction of disaster risk by better predicting impacts of El Niño and La Niña. Progress in Disaster Science, 2, 100022. https://doi.org/10.1016/j.pdisas.2019.100022
- Ham, Y.-G., Kug, J.-S., Park, J.-Y., & Jin, F.-F. (2013). Sea surface temperature in the north tropical Atlantic as a trigger for El Niño/Southern Oscillation events. *Nature Geoscience*, 6(2), 112–116. https://doi.org/10.1038/ngeo1686
- Harris, C. R., Millman, K. J., Van Der Walt, S. J., Gommers, R., Virtanen, P., Cournapeau, D., et al. (2020). Array programming with NumPy [Software]. Nature, 585, 357–362. https://doi.org/10.1038/s41586-020-2649-2
- Hoyer, S., Roos, M., Joseph, H., Magin, J., Cherian, D., Fitzgerald, C., et al. (2023). xarray [Software]. Zenodo. https://doi.org/10.5281/ ZENODO.7850711
- Hunter, J. D. (2007). Matplotlib: A 2D graphics environment [Software]. Computing in Science & Engineering, 9, 90–95. https://doi.org/10.1109/ MCSE.2007.55
- Iizumi, T., Luo, J.-J., Challinor, A. J., Sakurai, G., Yokozawa, M., Sakuma, H., et al. (2014). Impacts of El Niño Southern Oscillation on the global yields of major crops. *Nature Communications*, 5(1), 3712. https://doi.org/10.1038/ncomms4712
- Izumo, T., Vialard, J., Dayan, H., Lengaigne, M., & Suresh, I. (2016). A simple estimation of equatorial Pacific response from windstress to untangle Indian Ocean Dipole and Basin influences on El Niño. *Climate Dynamics*, 46(7–8), 2247–2268. https://doi.org/10.1007/s00382-015-2700-4
- Jiang, L., & Li, T. (2021). Impacts of tropical North Atlantic and equatorial Atlantic SST anomalies on ENSO. Journal of Climate, 1–58. https:// doi.org/10.1175/JCLI-D-20-0835.1
- Jiang, L., Li, T., & Ham, Y. (2022). Critical role of tropical North Atlantic SSTA in Boreal summer in affecting subsequent ENSO evolution. Geophysical Research Letters, 49(8), e2021GL097606. https://doi.org/10.1029/2021GL097606
- Jin, F.-F. (1997). An equatorial ocean recharge paradigm for ENSO. Part I: Conceptual model. *Journal of the Atmospheric Sciences*, 54(7), 19–829. https://doi.org/10.1175/1520-0469(1997)054<0811:aeorpf>2.0.co;2
- Khider, D., Emile-Geay, J., Zhu, F., James, A., Landers, J., Ratnakar, V., & Gil, Y. (2022). Pyleoclim: Paleoclimate timeseries analysis and visualization with Python [Software]. Paleoceanography and Paleoclimatology, 37. https://doi.org/10.1029/2022PA004509
- Li, X., Xie, S.-P., Gille, S. T., & Yoo, C. (2016). Atlantic-induced pan-tropical climate change over the past three decades. *Nature Climate Change*, 6(3), 275–279. https://doi.org/10.1038/nclimate2840
- McGregor, S., Stuecker, M. F., Kajtar, J. B., England, M. H., & Collins, M. (2018). Model tropical Atlantic biases underpin diminished Pacific decadal variability. *Nature Climate Change*, 8(6), 493–498. https://doi.org/10.1038/s41558-018-0163-4
- McGregor, S., Timmermann, A., Stuecker, M. F., England, M. H., Merrifield, M., Jin, F.-F., & Chikamoto, Y. (2014). Recent Walker circulation strengthening and Pacific cooling amplified by Atlantic warming. *Nature Climate Change*, 4(10), 888–892. https://doi.org/10.1038/ nclimate2330
- Naha, R., McGregor, S., & Singh, M. (2023). Exploring the symmetries of pantropical connections between the tropical Atlantic and Pacific basins. *Journal of Climate*, 36(15), 5251–5265. https://doi.org/10.1175/JCLI-D-22-0355.1
- Okumura, Y. M., & Deser, C. (2010). Asymmetry in the duration of El Niño and La Niña. Journal of Climate, 23(21), 5826–5843. https://doi.org/ 10.1175/2010JCL13592.1
- Planton, Y. Y., Guilyardi, E., Wittenberg, A. T., Lee, J., Gleckler, P. J., Bayr, T., et al. (2021). Evaluating climate models with the CLIVAR 2020 ENSO metrics package. Bulletin of the American Meteorological Society, 102(2), E193–E217. https://doi.org/10.1175/BAMS-D-19-0337.1
- Polo, I., Martin-Rey, M., Rodriguez-Fonseca, B., Kucharski, F., & Mechoso, C. R. (2015). Processes in the Pacific La Niña onset triggered by the Atlantic Niño. *Climate Dynamics*, 44(1–2), 115–131. https://doi.org/10.1007/s00382-014-2354-7
- Prichard, D., & Theiler, J. (1994). Generating surrogate data for time series with several simultaneously measured variables. *Physical Review Letters*, 73(7), 951–954. https://doi.org/10.1103/PhysRevLett.73.951
- Purich, A., Boschat, G., & Liguori, G. (2021). Assessing the impact of suppressing Southern Ocean SST variability in a coupled climate model. Scientific Reports, 11(1), 22069. https://doi.org/10.1038/s41598-021-01306-2
- Rayner, N. A., Parker, D. E., Horton, E. B., Folland, C. K., Alexander, L. V., Rowell, D. P., et al. (2003). Global analyses of sea surface temperature, sea ice, and night marine air temperature since the late nineteenth century [Dataset]. Journal of Geophysical Research, 108, 4407. https://doi.org/10.1029/2002JD002670
- Richter, I., Kosaka, Y., Kido, S., & Tokinaga, H. (2023). The tropical Atlantic as a negative feedback on ENSO. *Climate Dynamics*, 61(1–2), 309–327. https://doi.org/10.1007/s00382-022-06582-w
- Richter, I., Tokinaga, H., Kosaka, Y., Doi, T., & Kataoka, T. (2021). Revisiting the tropical Atlantic influence on El Niño–Southern Oscillation. Journal of Climate, 34(21), 8533–8548. https://doi.org/10.1175/JCLI-D-21-0088.1
- Rodríguez-Fonseca, B., Polo, I., García-Serrano, J., Losada, T., Mohino, E., Mechoso, C. R., & Kucharski, F. (2009). Are Atlantic Niños enhancing Pacific ENSO events in recent decades? *Geophysical Research Letters*, 36(20), L20705. https://doi.org/10.1029/2009GL040048
- Ruprich-Robert, Y., Moreno-Chamarro, E., Levine, X., Bellucci, A., Cassou, C., Castruccio, F., et al. (2021). Impacts of Atlantic multidecadal variability on the tropical Pacific: A multi-model study. *npj Climate and Atmospheric Science*, 4(1), 33. https://doi.org/10.1038/s41612-021-00188-5
- Ruprich-Robert, Y., Msadek, R., Castruccio, F., Yeager, S., Delworth, T., & Danabasoglu, G. (2017). Assessing the climate impacts of the observed Atlantic multidecadal variability using the GFDL CM2.1 and NCAR CESM1 global coupled models. *Journal of Climate*, 30(8), 2785–2810. https://doi.org/10.1175/JCL1-D-16-0127.1
- Smith, T. M., Reynolds, R. W., Peterson, T. C., & Lawrimore, J. (2008). Improvements to NOAA's historical merged land–ocean surface temperature analysis (1880–2006) [Dataset]. Journal of Climate, 21, 2283–2296. https://doi.org/10.1175/2007JCL12100.1

19448007,

- Theiler, J., Eubank, S., Longtin, A., Galdrikian, B., & Doyne Farmer, J. (1992). Testing for nonlinearity in time series: The method of surrogate data. *Physica D: Nonlinear Phenomena*, 58(1–4), 77–94. https://doi.org/10.1016/0167-2789(92)90102-S
- Williams, A. I. L., Jeevanjee, N., & Bloch-Johnson, J. (2023). Circus tents, convective thresholds, and the non-linear climate response to tropical SSTs. Geophysical Research Letters, 50(6), e2022GL101499. https://doi.org/10.1029/2022GL101499
- Zhang, W., Jiang, F., Stuecker, M. F., Jin, F.-F., & Timmermann, A. (2021). Spurious North tropical Atlantic precursors to El Niño. Nature Communications, 12(1), 3096. https://doi.org/10.1038/s41467-021-23411-6